Author's response to reviews

Title: Serum screening in first trimester to predict pre-eclampsia, small for gestational age and preterm delivery: Systematic review and meta-analysis

Authors:

Yan Zhong (azxyq731@163.com)
Fufan Zhu (yanmie137@126.com)
Yiling Ding (dylcsu@sina.cn)

Version: 2 Date: 17 December 2014

Author's response to reviews:

Numerous studies have shown that abnormal concentration of first trimester serum markers is related to the onset of preeclampsia, small for gestational age and preterm delivery. We investigated the accuracy of serum biochemical markers (Pregnancy- Associated Plasma Protein-A (PAPP-A), human Chorionic Gonadotropin (hCG), Placental Growth Factor (PIGF), Placental Protein 13 (PP13) used in first trimester serum screening in predicting preeclampsia, small for gestational age (SGA) and preterm delivery.